Author’s response to reviews

Title: Outcomes of endoscopic management of primary and refractory postcholecystectomy biliary leaks in a multicentre review of 178 patients

Authors:

Jorge Canena (jmtcanena@live.com.pt)
David Horta (david0rta@hotmail.com)
João Coimbra (jmgoimbra@sapo.pt)
Liliane Meireles (lilianeenailil@gmail.com)
Pedro Russo (pedro.me.russo@gmail.com)
Inês Marques (inesnmarques@zonmail.pt)
Leonel Ricardo (14ricardo@gmail.com)
Catarina Rodrigues (catarinagr@hotmail.com)
Tiago Capela (teixeira.capela@gmail.com)
Diana Carvalho (dianafbc.carvalho@gmail.com)
Rafaela Loureiro (loureiro.rafaela@gmail.com)
António M Dias (amateus.dias@netcabo.pt)
Gonçalo Ramos (goncalo.o.ramos@gmail.com)
António P Coutinho (avpc@ip.pt)
Carlos Romão (menezes_romao@yahoo.com)
Pedro M Veiga (geral@curva-de-gauss.pt)

Version: 2 Date: 30 March 2015

Author’s response to reviews:

As suggested
1) Line numbering and page numbers have been included
2) List of abbreviations have been included
3) Acknowledgments have been included
4) Disclosures was replaced by competing interests